Author’s response to reviews

Title: Nursing students motivation toward their studies - a survey study

Authors:

Kerstin E L Nilsson (kerstin.nilsson@fhs.gu.se)
Margareta I Warren Stomberg (margareta.warren.stomberg@his.se)

Version: 4 Date: 23 April 2008

Author’s response to reviews:

Dear Editor-in-chief;

Hereby, we submit the final revised manuscript: Nursing Students Motivation towards their Studies - a Survey Study. We hope we have handled the tables and figures in a correct way. We had some difficulties to interpret the instructions. If not correct please let us know.

On behalf of the authors

Kerstin Nilsson